Author's response to reviews

Title: Enhancing GPs' commitment in a hospital-based study. Rationale, design and participation rates in the Diagest 3 GP motivational study.

Authors:

Christophe Berkhout (christophe.berkhout@univ-lille2.fr)
Marie Vandaele-Bétancourt (marievandaele@yahoo.fr)
Stéphane Robert (stephane_r0b@yahoo.fr)
Solène Lespinasse (croqdoc@yahoo.fr)
Gamil Mitha (gamil.mitha@laposte.net)
Quentin Bradier (gbradier@yahoo.fr)
Anne Vambergue (anne.vambergue@chru-lille.fr)
Pierre Fontaine (pierre.fontaine@chru-lille.fr)

Version: 3 Date: 15 October 2011

Author's response to reviews:

The abstract has been modified to match instructions for authors, and formatted in 4 sections. The description of the methods was modified to be more explicit.

Details regarding ethical approval for the main study were specified in the introduction section, and complementary approval regarding the GP motivational study were mentionned at the end of the methods section.